Author's response to reviews

Title: A generic framework for web-based management of pseudonymized data

Authors:

Ronald Lautenschlaeger (ronald.lautenschlaeger@tum.de)
Florian Kohlmayer (florian.kohlmayer@tum.de)
Fabian Prasser (prasser@in.tum.de)
Claudia Eckert (claudia.eckert@in.tum.de)
Klaus A Kuhn (klaus.kuhn@lrz.tum.de)

Version: 2 Date: 22 May 2015

Author's response to reviews: see over
Dear Arlene Pura,

thank you for your comments on the manuscript. We have revised the document:

- Line numbers have been added.
- Page numbers have been added.
- Page breaks have been removed.

Please let me know if you have any further questions.

With my best regards,
Ronald Lautenschlaeger (corresponding author)
ronald.lautenschlaeger@tum.de